Author's response to reviews

Title: Impact of coronary artery calcification on risk assessment in asymptomatic subjects with family history of premature coronary artery disease

Authors:

Catalin Taraboanta (ctaraboanta@providencehealth.bc.ca)
Cameron Hague (cjhague75@gmail.com)
John G.B. Mancini (mancini@interchange.ubc.ca)
Bruce B. Forster (bruce.forster@vch.ca)
Jiri Frohlich (jifr@interchange.ubc.ca)

Version: 2 Date: 28 February 2012

Author's response to reviews: see over
Dear Miss Marinette Lacson

Please find attached the revised manuscript that now incorporates the changes that have been requested in your last correspondence.

1) Abstract is now 178 words that include the trial registration and it’s structured in three sections as per your request. Though understandable the use of acronyms are in fact facilitating the reading process once they’ve been defined.

2) A new competing interests section has been created and inserted after the conclusions section of the article.

3) Author’s contribution section has been edited and inserted right before the acknowledgements section of the paper.

Thank you for considering reviewing for publication of this manuscript.

Best regards,
Catalin Taraboanta